Author's response to reviews

Title: Community-associated Methicillin-resistant Staphylococcus aureus Bacteremia and Endocarditis among HIV Patients: a cohort study

Authors:

John P Furuno (furuno@ohsu.edu)
Jennifer K Johnson (jkjohnson@som.medicine.edu)
Marin L Schweizer (marin_schweizer@uiowa.edu)
Anayochukwu Uche (uchenayo@hotmail.com)
Oscar C Stine (ostin001@umaryland.edu)
Simone M Shurland (simone.shurland@fda.hhs.gov)
Graeme N Forrest (forrestg@ohsu.edu)

Version: 5 Date: 29 October 2011

Author's response to reviews:

Corrected 2 typos - Changed tidy to study in Discussion 3rd paragraph
Changed Capital V in hepatitis C virus in results